vc/2-Dec-03
Week 49 starting 01 dec 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

Today(2/12/03)
	Forward look,   & Next Magnet 

	TBA1
	NOT Active
	
	

	TBA2
	3021
	Connection
	

	TBC1
	1029
	To remove
	next ?

	TBC2
	SSS04
	Warming up
	Next sss

	TBD2
	Use 3059
	FOR COMMISSIONING
	

	TBE1
	
	 NEEDS REPAIR
	

	TBE2
	3070
	Connection + pumpdown
	

	TBF1
	3065
	4.4 k & Warm up
	Disconnect & bring new

	TBF2
	1038
	Connection + 
	


 In pipeline ?
Test Coordinator this week:  Marco B;  MTM  Tech. on call:  (16-3602) : 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. Hardware Recog. System implementation/use

2. Commissionind D Cluster

3. AOB

Bench Status ( Tues 2 dec at 1800 hrs)
	SM18 LHC Magnet Test Facility [image: image1]


	TBA1
	_
	_
	 


	TBA2
	MBAR3021
	Prep2.3 IAP @ warm LF
	1.9 K


	TBB1
	_
	_
	 


	TBB2
	_
	_
	 


	No records returned.


	TBC2
	SSS004
	PT12 Warm Up
	Warming


	TBD1
	_
	_
	 


	TBD2
	_
	_
	 


	TBE1
	-
	Not Activated
	 


	TBE2
	MBBL3070
	Prep2.3 IAP @ warm LF
	300 K


	TBF1
	MBAR3065
	PT12 Warm Up
	1.9 K


	TBF2
	MBBR1038
	New Magnet arrived
	300 K
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Problems of Last week

	Date
	Problem
	Type
	Application/

soft.
	Action/advise

	Tuesday 25_A


	TBC1: vapour appearing in the anticryostat after cleaning


	MTM
	
	Patrick called. Temperature checks. OK

Excessive alcohol used during the cleaning . 

	Tuesday 25_N


	TBC1: IAP @ cold : POTAIM, Qloc card amplitudes not adjustable. 


	MTM
	IAP
	

	Tuesday 25_N


	TBC1: After slow PA sunMTA20 hung 


	AB-CO
	
	

	Tuesday 25_N


	TBC1: Provoked Quench all HF

Server hung.
	AB/CO
	
	Shutdown of Sunmta19 and re-start.

 

	Tuesday 25_N


	TBC1: Training quench 1

No folder in MATEST for 1029
	MTM
	
	 Quench table stored in desktop of PC

	Tuesday 25_N

Wednesday 26_M

Sunday 30_morning


	TBC1: (3059) WP04 dielectric test step 4 not present

TBE1 (2027) WP04, WP07 dielectric test step 15 not present
	EST/ISS

AT/ADM
	MTF
	Solved from Monday 2/12.

 

	Wednesday 26_M


	TBE1: IAP @warm not done. Magnet already disconnected.

Two travellers for the magnet 2027?
	MTM
	
	 Authorization of disconnection signed by OP team but no checks that IAP was performed (no signature).

	Wednesday 26_M


	TBE2: Warm up : LF DAQ showing MXI move error in crate 1

DAQ not getting armed.
	MTM
	
	 Giorgio called.

Reset then OK

	Thursday 27_M


	TBF2: SPA. Not able to arm HF, LF acquisition.
	MTM
	
	

	Thursday 27_M


	TBF2: MM0 launched

Still CPU error.
	AB-CO
	
	 

	Thursday 27_M


	TBC1: Machine cycle: read current is 0 in aperture 1, already present during MM0 but equipment support not informed.
	MTM
	
	Cable of current reading for SSS removed. OK for the load-line.

Follow the procedure written in the grey folder MM. 

	Thursday 27_N


	TBF2: Launching hfoff messages “MP , MF’  not found. AQA shows faulty quench.
	MTM
	AQA?
	

	Thursday 27_N


	TBF2: Training quench 8: HF not armed at the start. Abort. Restart TEMA, ramp down the current.
	MTM
	
	 

	Thursday 27_N


	TBC1: Full load line. MMP hung when launching the test.
	AB/CO
	MMP
	TEMA closed and restart then OK.

	Friday 28_A


	TBF2: Cryo lost during training.
	ACR
	
	

	Friday 28_A


	TBE2: Switch to Tcoil position not possible automatically. Done it manually but when T-coil on X2 was launched the signal “Recording Armed “did not appear. But T-coil reading was possible.
	MTM
	
	

	Friday 28_A


	TBC1: reading of TT_821 erratic and wrong
	MTM
	
	

	Friday 28_N


	TBF2: Faulty quench. Vgas and Vext lead
	MTM
	
	Bad contact on the VT connector.

	Saturday 29_M


	TBE2: MM positions in ap 1.Coils from sector 1 to 4 seem to be broken.
	MTM
	
	Giorgio called

Connector at the rear of mobile rack not tied properly (from amplifier to fixed rack)

	Saturday 29_M


	TBE2: Provoked quench all HF (1.5 kA)

Signal not found in AQA 
	MTM
	
	

	Saturday 29_A


	TBE2:  hf-lf config files generated by HWREC are not correct 
	MTM
	HWREC
	Prepared again by Daniel and Giorgio.

	Saturday 29_A


	TBE2 (1032):  TT-821 not giving good T at cold 
	
	
	Non conformity opened by Daniel.

	Saturday 29_A

Sunday 30_A


	TBE2:  Quench a01 not sent to storage.

Same problem on TBC1, TBF1. 
	MTM
	
	No space in the disks?

	Saturday 29_N


	TBE2:  Full Load line. Some measurements not exported. Coil name invalid format
	MTM
	
	To be investigated.

	Sunday 30_M


	TBF1: SPA but interlock appears (cable for  temperature)
	MTM
	
	Giorgio re-mount the support for the cable temperature reading. Then OK

	Sunday 30_M


	TBF1: MM position, coil of ap 1 missing.

Cable between TRU and mobile rack should be repaired.
	MTM
	
	

	Sunday 30_A


	TBC1: No data available for quench. Workstation crashed

Message: No space left on for device.
	AB-CO
	
	Magnus Bjork : data will be saved on vol7.

	Mon 01_M


	TBC1: SSL at 4.4 K quench in Vclamp

Level of Helium in bus bar 60%: OK
	MTM
	
	

	Mon 01_M


	TBF1: Full load line. Problem with coupling between motor and shaft.
	MTM
	
	

	Mon 01_N


	TBF1: DRQ at 50 A/s in theVbus ext.

Level of Helium in bus bar 60%: OK
	MTM
	
	


